Author's response to reviews

Title: Disease knowledge level is a noteworthy determinant of anxiety and depression in patients with Chronic Obstructive Pulmonary Disease: a cross-sectional studies

Authors:

Qiao Zhang (zhangqiaoo@gmail.com)
Jiangrong Liao (liaojiangrong@yeah.net)
Xiuqing Liao (XQ_Liao@126.com)
Xiuling Wu (xl_wu2001@163.com)
Min Wan (W_an521@sina.com)
Changzheng Wang (czwang@yeah.net)
Qianli Ma (cqmql@163.com)

Version: 2 Date: 13 January 2014

Author's response to reviews:

Dear Editor:

Thanks for your carefully checking of my manuscript. It's so sorry that some important information had been ignored by us. Here, all changes are listed one by one.

1. Insert the section title "Conclusion" before the last paragraph of "Discuss" section:

In the original version of the manuscript, we ignored the section title "Conclusion";

2. Add "Acknowledgements" section:

We misunderstand the "Instructions for authors". The "Acknowledgements" section had been seem as an optional section. Now, we added this section.

Thank you very much for consideration!

Sincerely Yours,

Qianli Ma

Institute of Respiratory Diseases, Chronic airway disease care center
Xinqiao Hospital,
Third Military Medical University,
Chongqing 400037, China